Author’s response to reviews

Title: Evaluation of health promotion in schools: a realistic evaluation approach using mixed methods

Authors:

Jeanine Pommier (Jeanine.Pommier@ehesp.fr)
Marie-Renée Guevel (Marie-Renee.Guevel@ehesp.fr)
Didier Jourdan (Didier.Jourdan@univ-bpclermont.fr)

Version: 2 Date: 11 January 2010

Author’s response to reviews:

Dear Editor,

To answer your comment on the name of the ethics committee (send by e-mail Thursday 7th, 2010), we added the following sentence p.12 of our manuscript in the first paragraph of the data collection section: “The questionnaires were registered at a national ethics committee, the “Commission nationale de l’informatique et des libertés” (CNIL), the national board in charge of data protection and liberties (http://www.cnil.fr/english/the-cnil/).”

Thank you for your time and consideration.

Yours sincerely,

The authors

Jeanine Pommier, Marie-Renée Guevel, Didier Jourdan